%  $BI=Bj(B  $BB4O@(B!!
%
%  $BMzNr(B  2002/02/22 $B;3ED(B $BM35.;R(B 
%  
%
%%% style set %%%
\documentclass[a4j,12pt,openbib,titlepage]{jarticle}
\usepackage{ascmac}
\usepackage{./styfile/pageset}
\usepackage{Dennou6}
\usepackage{./styfile/jmsj}
\pagestyle{DAheadings}

%% $B%?%$%H%k@_Dj(B
\Dtitle[]{$B@VF;GH$N@~7AM}O@(B \\
                         Linear theory of equatorial waves} 
\Dauthor[]{$B;3ED(B $BM35.;R(B \\ YAMADA Yukiko \\
 $BKL3$F;Bg3XM}3XItCO5e2J3X2J(B  $BCO5eN.BNNO3X8&5f<<(B}
 
\Ddate[]{2002 $BG/(B 1 $B7n(B 31 $BF|(B}
\Dfile{}


%%% text start %%%
\begin{document}

%% $B%?%$%H%k(B 
\maketitle

%% $B35MW(B
\begin{abstract}
\Dparskip
\input{./texs/abst.tex}
\end{abstract}
\pagebreak

%% $BL\<!(B
\def\thepage{\arabic{page}}
\tableofcontents
\listoffigures
\pagebreak

%% $BK\J8(B
\setcounter{page}{1}
\Dparskip
\input{./texs/chap1-intro.tex}
\pagebreak
\input{./texs/chap2-model.tex}
\pagebreak
\input{./texs/chap3-mode.tex}
\pagebreak
\input{./texs/chap3-modefigure.tex}
\pagebreak
\input{./texs/chap4-nummodel.tex}
\pagebreak
\input{./texs/chap5-discuss.tex}
\pagebreak
\input{./texs/chap6-result.tex}

%% $B;29MJ88%(B
\pagebreak 
\bibliographystyle{jas}
\bibliography{./REF}

\nocite{ispack}
\nocite{dcl}
\nocite{tsuron}
\nocite{toku}
\nocite{kikouhendou}

\end{document}

























